Author's response to reviews

Title: epiPATH: an information system for the storage and management of molecular epidemiology data from infectious pathogens

Authors:

Alicia Amadoz (alicia.amadoz@uv.es)
Fernando Gonzalez-Candelas (fernando.gonzalez@uv.es)

Version: 2 Date: 13 November 2006

Author's response to reviews:

We have included detailed instructions for the reviewers accessing and/or downloading the software described in the manuscript while keeping their anonymity. Although it is already stated in the manuscript, we would like to confirm that the software will be available for non-commercial usage under a GNU Public License and through our own web server and an additional public repository such as sourceforge. We hope that with these clarifications the paper is now ready for peer evaluation.

Best regards,

Fernando Gonzalez-Candelas.